Author’s response to reviews

Title: Cognitive Health Outcomes Investigation of the Comparative Effectiveness of Dementia Screening: The Indiana University CHOICE Study

Authors:

Nicole R Fowler (fowlemr@upmc.edu)
Amanda Harrawood (aharrawo@umail.iu.edu)
Amie Frame (amiframe@iupui.edu)
Anthony J Perkins (antperki@iupui.edu)
Sujuan Gao (sgao@iu.edu)
Christopher M Callahan (ccallaha@iupui.edu)
Geg A Sachs (sachsg@iu.edu)
Dustin D French (dustin.french@northwestern.edu)
Malaz A Boustani (mboustan@iupui.edu)

Version: 2
Date: 24 April 2014

Author’s response to reviews:

We have made the formatting changes as indicated in an email sent to us from Ms. McGavin on April 9th.